Author's response to reviews

Title: Three-dimensional quantitative analysis of adhesive remnants and enamel loss resulting from debonding orthodontic molar tubes

Authors:

Joanna Janiszewska-Olszowska (jjo@pum.edu.pl)
Katarzyna Tandecka (katarzyna.tandecka@tu.koszalin.pl)
Tomasz Szatkiewicz (tomasz.szatkiewicz@tu.koszalin.pl)
Katarzyna Sporniak-Tutak (katarzyna@aestheticdent.pl)
Katarzyna Grocholewicz (kgrocholewicz@o2.pl)

Version: 3  
Date: 14 August 2014

Author's response to reviews:

Dear Sirs,

The authors thank for all the comments and corrections. We have done our best to introduce all of them, as proposed.

„Brackets“ have been replaced by „molar tubes“ and unclear sentences have been rephrased as kindly suggested. Figures 2, 3 and 4 have been combined into a single figure as proposed. Figure numbering has been adjusted. A reference has been added to the first paragraph of Discussion section as suggested.

Information on calibration procedure and on the lack of error study has been added to the Material and Methods section as proposed. Study sample size appropriateness has been verified as suggested. Conclusions have been corrected according to the suggestion.

The authors are very grateful for the all the wishful remarks, as they improve the manuscript quality. We would be very proud to have our article published in Head and Face Medicine.

Yours sincerely,

Joanna Janiszewska-Olszowska